Author's response to reviews

Title: Comparisons of Intensity-Duration Patterns of Physical Activity in the US, Jamaica and 3 African Countries

Authors:

Lara R Dugas (ldugas@luc.edu)
Pascal Bovet (pascal.bovet@chuv.ch)
Terrence E Forrester (terrence.forrester@uwimona.edu jm)
Estelle V Lambert (vicki.lambert@uct.ac.za)
Jacob Plange-Rhule (jprhule@gmail.com)
Ramon A Durazo-Arvizu (rdurazo@luc.edu)
David A Shoham (dshoham@luc.edu)
Jacolene Kroff (jacolen.kroff@uct.ac.za)
Guichan Cao (gcao@luc.edu)
Richard S Cooper (rcooper@luc.edu)
Soren Brage (soren.brage@mrc-epid.cam.ac.uk)
Ulf Ekelund (Ulf.Ekelund@mrc-epid.cam.ac.uk)
Amy Luke (aluke@luc.edu)

Version: 2 Date: 25 April 2014

Author's response to reviews:

25th April, 2014

Diana Marshall
Senior Managing Editor
BMC Public Health

Dear Dr. Marshall

I am re-submitting a manuscript titled “Comparisons of Intensity-Duration Patterns of Physical Activity in the US, Jamaica and 3 African Countries” for consideration of publication in BMC Public Health. I have added the email address for all the corresponding authors and have added line numbers.

Thank you for your consideration.

Yours truly,

Lara Dugas, PhD, MPH